Author's response to reviews

**Title:** Time Trends of Baseline Demographics and Clinical Characteristics of HIV infected Children Enrolled in Care and Treatment Service in Dar es Salaam, Tanzania

**Authors:**

- david M sando (dsando.tz@gmail.com)
- Donna Spiegelman (stdls@hsph.harvard.edu)
- Lameck Machumi (lmachumi@mdh-tz.org)
- Mary Mwanyika-Sando (mmwanyika.sando@gmail.com)
- Eric Aris (aeric@mdh-tz.org)
- Aisa Muya (aisamuya@gmail.com)
- Elizabeth Jackson (ejackson@hsph.harvard.edu)
- Till Baernighausen (tbaernig@hsph.harvard.edu)
- Ellen Hertzmark (stleh@channing.harvard.edu)
- Guerino Chalamila (gchalamilla@gmail.com)
- Wafaie Fawzi (mina@hsph.harvard.edu)

**Version:** 4  **Date:** 19 February 2015

**Author's response to reviews:** see over
Changes made in the manuscript
"You have to change mean age in the abstract as well (results and conclusion). There is a typo using NV instead of NVP in the text. Better if you change it into nevirapine. Likewise please spell out the antiretroviral drug names entirely."

I have changed the median age in the abstract and conclusion to mean age. I have also corrected the typo NV to NVP and have spell out the antiretroviral drug names entirely.

Requesting consent statement:

Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian

I have added the sentence in the method section: written consent was obtained from parents or guardians of the children involved in the study.